Author's response to reviews

Title: Low expression of TFPI-2 associated with poor survival outcome in patients with breast cancer

Authors:

Cheng Xu (xucheng@live.cn)
Huijun Wang (huijunwang@fudan.edu.cn)
Hongyu He (hehongyu@live.cn)
Fengyun Zheng (zhengfengyun@gmail.com)
Yating Chen (yatingch99@hotmail.com)
Jin Zhang (jeanzhj@gmail.com)
Xiaoyan Lin (linxiaoyan077@163.com)
Duan Ma (duanma@shmu.edu.cn)
Hongwei Zhang (zhang.hongwei@zs-hospital.sh.cn)

Version: 5 Date: 28 January 2013

Author's response to reviews:

Dear Editor,

In this revisions on 28 January 2013, I've delete the term "to determine whether the independent prognostic values of predict factors are statistically significant" on page 2 (Abstract) as the Editor indicates. The same change has been made in manuscript details.

With best wishes,

Cheng Xu